Author's response to reviews

Title: Towards the construction of health workforce metrics for Latin America and the Caribbean

Authors:

Gustavo G Nigenda JR, I (gnigenda@insp.mx)
Maria H Machado (helena.machado@saude.gov.br)
Fernando F Ruiz (fruiz@javeriana.edu.co)
Victor V Carrasco (victor.carrasco@upch.pe)
Patricia P Moline (pmoline@copresida.gob.do)
Sabajo S Girardi (gerardis@medicina.ufmg.br)

Version: 2 Date: 24 March 2011

Author's response to reviews: see over
Editors
Human Resources for Health Journal

Dear Editors:

We are sending attached a new version of the manuscript entitled "Towards the construction of health workforce metrics for Latin America and the Caribbean" for consideration of publication in this Journal containing the suggestions of two reviewers.

Changes suggested by the first reviewer include:
   a) Re organization of the sections in the document to be presented as a case study
   b) Re definition of the objective of the document
   c) Exclusion of paragraphs containing irrelevant information
   d) Accurate information about the literature search
   e) Correction of all minor essential revisions

For the second reviewer modifications were:
   a) Changes in the English language phrasing.
   b) Corrections in references numbering

We look forward to the results of our submission.

Sincerely